Author's response to reviews

Title: Use of health systems and policy research evidence in the health policymaking in eastern Mediterranean countries: views and practices of researchers

Authors:

Fadi El-Jardali (fe08@aub.edu.lb)
John N Lavis (lavisj@mcmaster.ca)
Nour Ataya (na92@aub.edu.lb)
Diana Jamal (dj06@aub.edu.lb)

Version: 6 Date: 10 January 2012

Author's response to reviews:

Dear Editorial Team,

Thank you for your e-mail.

We made sure to double check the order of the tables in the text. Kindly find the corrected version of the manuscript enclosed.

Sincerely,

Fadi

Fadi El-Jardali, MPH, PhD
Associate Professor of Health Management and Policy, American University of Beirut
Investigator, Program in Policy and Decision Making, McMaster University
Evidence to Policy Fellow, Alliance for Health Policy and Systems Research